Author's response to reviews

Title: Obesity and pregnancy: a transversal study from a low-risk maternity

Authors:

Ana Carolina S Calderon (ana_carol1@hotmail.com)
Silvana M Quintana (quintana@fmrp.usp.br)
Alessandra C Marcolin (dralemar@uol.com.br)
Aderson T Berezovski (atberezoz@fmrp.usp.br)
Luiz Gustavo O Brito (lgobrito@gmail.com)
Geraldo Duarte (gduarte@fmrp.usp.br)
Ricardo C Cavalli (rcavalli@fmrp.usp.br)

Version: 2 Date: 25 April 2014

Author's response to reviews: see over
It is with pleasure that we are re-sending a manuscript (Obesity and pregnancy: a transversal study from a low-risk maternity) for the analysis of your editorial board, intending to publish it in your respectful journal. This manuscript was previously analyzed by your peer-reviewers (MS: 2765388671202516) and final decision was reject, however, you gave the possibility to re-send it after careful and extensive review.

We have done the manuscript review and are resending it to you. We understand it may take much time than a standard period; we would appreciate if you could preview this for us. All modifications are highlighted with a blue mark. All reviewers´ comments are answered in a separate file in a red mark. We think it would increase discussions about the impact of obesity in obstetrical outcomes.

We certify that this manuscript is original, and there is no conflict of interests. We transfer all rights of this article to your journal in case of acceptance.

Sincerely,

Luiz Gustavo Oliveira Brito (on behalf of all authors)